Author's response to reviews

Title: Association of HFE common mutations with Parkinsons disease, Alzheimers disease and mild cognitive impairment in a Portuguese cohort

Authors:

Rita J Guerreiro (ritajoao@gmail.com)
Jose M Bras (brasjm@gmail.com)
Isabel Santana (isabelsantana@huc.min-saude.pt)
Cristina Januario (cristinajanuario@gmail.com)
Beatriz Santiago (ubaldo2500@hotmail.com)
Ana S Morgadinho (anamorgadinho@portugalmail.pt)
Maria H Ribeiro (mhgarrucho@huc.min-saude.pt)
John Hardy (hardyj@mail.nih.gov)
Andrew Singleton (singleta@mail.nih.gov)
Catarina R Oliveira (catarina@cnc.cj.uc.pt)

Version: 4 Date: 2 July 2006

Author's response to reviews:

Changes were made according to the Editorial Production Team suggestions.